Author's response to reviews

Title: Antinociceptive effect of ethanolic extract of Selaginella convoluta in mice

Authors:

Pedro GS de Sá (pedro_cdb@hotmail.com)
Xirley P Nunes (xirleypn@gmail.com)
Julianeli T de Lima (julianeliti@gmail.com)
José A de Siqueira-Filho (jose.siqueira@univasf.edu.br)
André P Fontana (andrepavioii@yahoo.com.br)
Jullyana S Siqueira (jullyanas@yahoo.com.br)
Lucindo J Quintans-Júnior (lucindo Jr@yahoo.com.br)
Patrícia KF Damasceno (patriciakauanna@hotmail.com)
Carla RC Branco (carlacardoso@uefs.br)
Alexsandro Branco (branco@uefs.br)
Jackson RGS Almeida (jackson.guedes@univasf.edu.br)

Version: 6 Date: 17 August 2012

Author's response to reviews: see over
Dear Editor,

Here are my thoughts about changes in the article, in order to improve the quality of it for publication. All changes made in the work and some languages corrections are highlighted in blue.

Referee #2

Reviewer's report: the authors have made significant improvements to their manuscript.

Quality of written English: Needs some language corrections before being published. Corrections were made in written language by a native and are in blue at the text.

After these considerations, the editor may consider the article for publication.

Best regards.

Dr. Jackson Roberto Guedes da Silva Almeida
Corresponding author